Author’s response to reviews

Title: Non-Adherence to the Single Dose Nevirapine Regimen for the Prevention of Mother-To-Child Transmission of HIV in Bindura Town, Zimbabwe: A cross-sectional analytic study.

Authors:

Lazarus R Kuonza (lkuonza@gmail.com)
Clemence D Tshuma (ctshuma@healthnet.zw)
Gerald N Shambira (gshambira@healthnet.org.zw)
Mufuta Tshimanga (mtshimang@ecoweb.co.zw)

Version: 3 Date: 7 April 2010

Author’s response to reviews: see over
Response to viewers’ comments:

Overall changes:

We have addressed the important issues that were highlighted by the peer reviewers in this re-submission.

In the methods section we revised the statement about ethical approval, as suggested by the editorial team.

We also made a few minor spelling corrections as highlighted in the document.

Reviewer 1: Elizabeth M. Stringer

No issues were raised by this reviewer.

Reviewer 2: Fiona Smaill

1. It is unclear how many HIV positive women did not attend the postpartum visit

   *In the first paragraph of the results we inserted the statement ‘Of the 273 PMTCT mothers who were due to attend the six-week postnatal care in the participating health institutions during the study period, 229 (83.9%) attended the postpartum visit.’ to address this concern.*

Reviewer 3: David Coetzee

No major concerns were raised by this reviewer